Author’s response to reviews

Title: Changes in mental disorder prevalence among conflict-affected populations: a prospective study in Sri Lanka (COMRAID-R)

Authors:

Chesmal Siriwardhana (chesmal@gmail.com)
Anushka Adikari (anushka.adikari@yahoo.com)
Gayani Pannala (gayanipannala@yahoo.com)
Bayard Roberts (bayard.roberts@lshtm.ac.uk)
Sisira Siribaddana (sisira.siribaddana@gmail.com)
Melanie Abas (melanie.abas@kcl.ac.uk)
Athula Sumathipala (athula.sumathipala@kcl.ac.uk)
Robert Stewart (robert.stewart@kcl.ac.uk)

Version: 2
Date: 2 September 2014

Author’s response to reviews:

Line numbers provided and figure removed form the main text.